Author's response to reviews

Title: Systematic review of clinical practice guidelines recommendations about primary cardiovascular disease prevention for older adults.

Authors:

Jesse Jansen (jesse.jansen@sydney.edu.au)
Shannon McKinn (shannon.mckinn@sydney.edu.au)
Carissa Bonner (carissa.bonner@sydney.edu.au)
Les Irwig (les.irwig@sydney.edu.au)
Jenny Doust (j.doust@bond.edu.au)
Paul Glasziou (pglaszio@bond.edu.au)
Brooke Nickel (brooke.nickel@sydney.edu.au)
Barbara van Munster (b.c.vanmunster@amc.uva.nl)
Kirsten McCaffery (kirsten.mccaffery@sydney.edu.au)

Version: 2
Date: 12 February 2015

Author's response to reviews:

As requested, we have now included a list of abbreviations, and a list of figure titles and legends at the end of the manuscript text.

Kind regards,
Shannon McKinn